Author's response to reviews

Title: Home-based music therapy - A systematic overview of settings and conditions for an innovative service in healthcare.

Authors:

   Wolfgang Schmid (Wolfgang.Schmid@dzne.de)
   Thomas Ostermann (thomaso@uni-wh.de)

Version: 5 Date: 12 October 2010

Author's response to reviews:

Dear Production team,
all major and minor formatting changes have been done as requested.
Kind regards
Thomas Ostermann